Dear Editor Lian-Sheng Ma and Reviewers:

Manuscript NO.: 76912, Minireviews

Thanks to Reviewers for the time and effort spent reviewing the manuscript. We have tried our best to improve the manuscript and have modified all issues that are raised in the peer-review reports.

Responds to Reviewers' comments:

Reviewer #1

- The paper may benefit if reframed into a more rigorous review format, e.g. systematic review of all treatment approaches for ADHD and the monitoring of their efficacy by means of functional MRI. However if the authors prefer to adhere to the current format, they should clearly state it is as "selective review", describe their search strategy and methodology and increase the number of references to >80.

- Response: We appreciate it very much for this good suggestion, and we have made changes as far as possible. We described the nature of our review and fully demonstrated our search strategy and methodology(from line153 to line172). Moreover, we have cited enough literature and enriched essential content(from line283 to line286).

Reviewer #2

- Authors are encouraged to double-check the reference form.

- Response: Thanks to Reviewer for the recognition. We have carefully checked the reference form.

Reviewer #3

- Recommend to add information on available data on efficacy and tolerability of non-stimulant options as recently approved viloxazine focuses more on comorbidities due to activity on serotonin receptors.

- Response: We appreciate it for the sincere advice, and we have done it according to your
ideas. The characteristics of viloxazine are briefly introduced.

Thanks again for your consideration of this manuscript.

Sincerely yours,
Xinpeng Xu

dr_xpxu@163.com

Chunfeng Xiao

springwindxiao@163.com